Author's response to reviews

Title: Effectiveness of the Austrian Disease-Management-Programme for Type 2 Diabetes: Study Protocol of a Cluster-Randomized Controlled Trial

Authors:

Andreas C. Soennichsen (andreas.soennichsen@pmu.ac.at)
Andreas Rinnerberger (andreas.rinnerberger@pmu.ac.at)
Maria G. Url (maria.url@gmx.at)
Henrike Winkler (henrike.winkler@pmu.ac.at)
Peter Kowatsch (p.kowatsch@pobox.com)
Gert Klima (gert.klima@stgkk.at)
Bernhard Fuerthauer (f.bernie@myhealth.at)
Raimund Weitgasser (r.weitgasser@salk.at)

Version: 3 Date: 29 May 2008

Author's response to reviews:

To the
Editorial Board of “Trials”
Salzburg, May 29th, 2008

Manuscript: Effectiveness of the Austrian Disease-Management-Programme for NIDDM: Study Proto-col of a Cluster-Randomized Controlled Trial, MS: 1454113065182510

Dear Editors,

We are happy that our manuscript has been accepted for publication in “Trials”. According to your requirements we made the following final changes:
1. We removed all capitals from the title
2. We corrected the author’s list
3. We removed the page numbers
4. We inserted the subheading “Conclusions” before the last two sentences of the discussion
5. We corrected the author’s contributions according to your suggestion
6. We included the statistician we consulted in the acknowledgements
7. We put all the reference numbers in the text in brackets
8. We omitted the “et al.” and included all missing authors in the references.

We hope that our changes comply with your requirements. Please feel free to contact us in case of any further questions.
Sincerely yours,
Andreas Sönnichsen